Technische Universität München Fakultät für Informatik

Dr. K. N. Verma verma@in.tum.de Room: MI 02.07.041

Language Based Security

Summer Semester 2006

7. Homework 28 June 2006

Exercise 1:

Modify the runClass class discussed in the lecture to load a given class from a given url using the URL class loader, and run the main method of that class.

a) Modify the class now by adding a policy stating which classes loaded from which urls have read, write, delete and execute permissions on local files.

b) State this using a policy file instead of coding it in the class definition.